Author's response to reviews

Title: Home-based HIV testing for men preferred over clinic-based testing by pregnant women and their male partners, a nested cross-sectional study

Authors:

Alfred O Osoti (alfredos@uw.edu)
Grace John-Stewart (gjohn@uw.edu)
James N Kiarie (jkiarie@swiftkenya.com)
Richardson Barbra (barbrar@u.washington.edu)
Kinuthia John (jkinuthia@uw.edu)
Daisy Krakowiak (daisykr@uw.edu)
Carey Farquhar (cfarq@uw.edu)

Version: 3 Date: 30 April 2015

Author's response to reviews:

Title Page: We have included this with the manuscript and uploaded.

2. Line Numbering: We have added the page numbers.

3. Competing Interests: We have added that there are no competing interests and we have no financial disclosures.

4. Authors' Contributions: We have specified the roles of the authors.

5. Acknowledgements: We have also written acknowledgments

6. List of abbreviations: These have been defined when used.